Author’s response to reviews

Title: European network for promoting the physical health of residents in psychiatric and social care facilities (HELPS): background, aims and methods

Authors:

Prisca Weiser (prisca.weiser@uni-ulm.de)
Thomas Becker (t.becker@uni-ulm.de)
Carolin Losert (carolin.losert@bkh-guenzburg.de)
Köksal Alptekin (koksal.alptekin@deu.edu.tr)
Loretta Berti (loretta.berti@univr.it)
Lorenzo Burti (lorenzo.burti@univr.it)
Alexandra Burton (a.burton@qmul.ac.uk)
Mocja Dernovsek (mocja-zvezdana.dernovsek@guest.arnes.si)
Eva Dragomirecka (dragonirecka@pcp.if3.cuni.cz)
Marion Freidl (marion.freidl@meduniwien.ac.at)
Fabian Friedrich (fabian.friedrich@meduniwien.ac.at)
Aneta Genova (aneta@bghelsinki.org)
Arunas Germanavicius (arunas.germanavicius@vrc.vu.lt)
Ulas Halis (halisulas@yahoo.com)
John Henderson (mac53@talktalk.net)
Peter Hjorth (peter.hjorth@rn.dk)
Taavi Lai (taavi.lai@ut.ee)
Jens Ivar Larsen (jil@rn.dk)
Katarzyna Lech (gabinet@harmony.pl)
Ramona Lucas (Ramona.Lucas@uab.cat)
Roxana Marginean (marginean_roxana@yahoo.com)
David McDaid (D.Mcdaid@lse.ac.uk)
Maya Mladenova (mmladenova@datacom.bg)
Povl Munk-Jørgensen (pmj@rn.dk)
Alexandru Paziuc (apaziuc@yahoo.com)
Petronela Paziuc (elapaziuc@yahoo.com)
Stefan Priest (s.priest@qmul.ac.uk)
Katarzyna Prot-Klinger (k.prot@wp.pl)
Johannes Wancata (johannes.wancata@meduniwien.ac.at)
Reinhold Kilian (reinhold.kilian@uni-ulm.de)

Version: 4 Date: 14 August 2009

Author’s response to reviews: see over
Dear BioMed Central Editorial Production Team:

Thank you for your advices concerning the HELPS manuscript. We were delighted to learn that our paper has been accepted for publication in BMC Public Health.

We have revised the manuscript: in the following please find a description of the final changes:

1. The Abstract is now subdivided into Background, Methods, Results, and Conclusions.
2. The reference list now contains all details; minor mistakes are corrected.
3. Affiliations are now in the following format: (department), institute, city, and country.
4. We added a Conclusions section following the Discussion.

Furthermore we checked the revised manuscript with regard to the journal style of BMC Public Health and corrected where necessary.

We hope the revised HELPS manuscript meet your demands. Please don’t hesitate to contact me if there are further questions or advices regarding the HELPS manuscript.

Thanks and respectfully yours, also on behalf of the co-authors,

Prisca Weiser